Future Science

Alzheimer's disease patient groups derived from a multivariate analysis of cognitive test outcomes in the Coalition Against Major Diseases dataset

Aim: The mini-mental state examination, commonly used to measure cognitive impairment of Alzheimer's disease (AD) patients, consists of five test categories. The final score is calculated as their total sum, implying a loss of information. Materials & methods: In this study, we propose a new multivariate approach to address this issue. Results: We analyzed the current largest AD-related coalition against major diseases dataset comprising 3717 patients of interest. Our clustering approach revealed five groups of patients associated with distinct characteristics and prognosis. Interestingly, only three cognitive test categories significantly contribute to their determination: registration, attention and recall. Conclusion: The insight that only these categories are critical for AD group determination may help to resolve the patients' educational background issue often discussed in relation to the mini-mental state examination assessment.

Lay abstract: The mini-mental state examination is commonly used to measure cognitive impairment of Alzheimer's disease (AD)-diagnosed patients. Traditionally, the resulting score of this test is assessed as the total number of correct answers to five types of questions. In this study, we employed a more sophisticated multivariate approach to determine the most common groups of AD patients defined by their cognitive performance in each single category separately. Interestingly, our results revealed that out of the currently employed five question categories only three are crucial for the patient group determination. This insight may impact on the cognitive test design in AD.

First draft submitted: 2 June 2016; Accepted for publication: 21 July 2016; Published online: 19 August 2016

Keywords: Alzheimer's disease • mini-mental state examination • multivariate analysis

Alzheimer's disease (AD) is the most common form of dementia, accounting for 60–80% of all cases [1]. It is currently considered incurable and it eventually leads to death. The progression of this disease is mostly onedirectional with the average survival times after diagnosis lying around 7 years [2], and the probability of living longer than 14 years is smaller than 3% [3].

The recognized main risk factors of AD comprise, but are not limited to, high age, mild cognitive impairment, lack of social engagement, low level of education, family history, *APOE* allele e4 genotype, and cardiovascular disease and traumatic brain injury [1]. Although the age has been recognized as the strongest risk factor, alone it is not sufficient to cause the disease. A mild cognitive impairment can also be associated with AD, however, it does not always lead to this neurological disease and individuals can return to a normal condition instead [1,4-6]. Multiple studies have previously suggested that social and cognitive engagement sup-

Inna Tishchenko^{1,2}, Carlos Riveros^{1,3}, Pablo Moscato*^{1,2} & Coalition Against Major Diseases

¹Information-Based Medicine Program, Hunter Medical Research Institute, New Lambton Heights, NSW, Australia ²School of Electrical Engineering & Computer Science, The University of Newcastle, Callaghan, NSW, Australia ³CReDITSS Unit, Hunter Medical Research Institute, New Lambton Heights, NSW, Australia *Author for correspondence:

^Author for correspondence: pablo.moscato@newcastle.edu.au



ports brain health and leads to a reduced risk of AD [7-9]. Besides these environmental factors, the polymorphic allele e4 of *APOE*, a major cholesterol carrier that supports injury repair in the brain, has been widely associated with AD, while the e2 allele is considered to decrease the risk [10-12].

In the course of AD progression, brain synapses start to fail transferring information and neurons eventually die [1]. Consequently, first symptoms of this disease include difficulties in remembering recent events, speaking, writing, planning, solving tasks and a decreased judgment [6]. Advanced stages of AD are associated with confusion, mood swings, depressions, long-term memory losses and disability to reason clearly [13].

Diagnosis of AD is usually based on patients' cognitive test results, medical and family history, and brain scans [1]. Although the mental state tests alone are not sufficient to determine or diagnose AD, they are essential for the evaluation of the disease progression stage. One of the most commonly applied tests is the 30-point mini-mental state examination (MMSE) questionnaire introduced in 1975 [14-16]. In 2010, a second edition of this questionnaire was released in ten different languages to enable its worldwide application [17]. The standard MMSE test consists of five categories: orientation, registration, attention, recall and language, and the number of maximal points per category is equal to 10, 3, 5, 3 and 9, respectively [14,16,18]. In this test, patients are first asked ten questions related to orientation in time and place. A registration task follows these questions and it includes learning and repetition of three object names. Next, patients are asked to spell backward a five-letter word to test their attention abilities. The recalling skills are determined by patients' capability to repeat the names of three previously learned objects. To assess patients' linguistic skills, they are instructed to name shown objects and write down a sentence. The final MMSE score is calculated as the total sum of all points, and it ranges between 0 and 30. Consequently, the test categories orientation, registration, attention, recall and language contribute to this score with the relative weights of 33.3, 10, 16.7, 10 and 30%, respectively. In practice, a total score of 27 or more is considered to represent a normal condition; scores below this threshold are associated with a mild (19-24 points), moderate (10-18 points) or severe $(\leq 9 \text{ points})$ cognitive impairment [19].

Along with the original MMSE version, there are several modifications occasionally applied in practice [20]. One of them incorporates three recall trials instead of one [21]; another version, the '26-point telephone MMSE', has been designed to be conducted over the phone and it skips four questions [22]. The 'modified mini-mental state (3MS) examination', on the other hand, has been developed to increase the test sensitivity to orientation, verbal fluency and ability to identify relations between objects [23]. While these cognitive test modifications employ different weights on similar categories, their results interpretation occurs in the same way as for the MMSE, where the total single score is computed as the sum of all points achieved in distinct categories.

Although the current evaluation of cognitive test results using a single score may appear convenient, its interpretation is rather inconclusive with regards to particular cognitive functions. Differences in weights (or the numbers of questions) assigned to single test categories affect the final score calculation; for instance, in the standard MMSE test the orientation skills affect the calculation of the final score the most, while the recalling and registration abilities only provide limited contributions. Educational background has also been previously shown to affect the total MMSE scores, and a revision or adjustment of the current standard MMSE test assessment has been recommended [24-29]. Thus, employing different scoring systems, previously described as test modifications, lead to varying final results incompatible for a direct comparison across test versions and patients.

In this study, we aim at addressing the cognitive test single total score-induced bias by identifying patient groups emerging from a multivariate analysis of the cognitive test results recorded in the current largest AD-related coalition against major diseases (CAMD) database. This project is motivated by the observation that the CAMD database is an aggregation of several trials that employed varying cognitive test versions and are therefore generally incompatible or not directly comparable among each other in terms of the calculated patients' overall cognitive performance. Thus, our goal is to resolve which test categories contribute to the patient group differentiation the most and to build a model that can be employed to reliably assign a patient to an AD cognitive group associated with certain impairment characteristics and clinical prognosis, independent of the questionnaire version. Furthermore, our objective is to discuss the relation between the determined classification model and patients' educational background.

Materials & methods Dataset description

Toward achieving the objectives defined above, we analyzed the CAMD dataset based on 6500 ADdiagnosed patients and 24 clinical trials [30]. CAMD was formed in 2008 by the Critical Path Institute, in collaboration with the Engelberg Center for Health Care Reform at the Brookings Institution (Washington, DC, USA). The Coalition brings together patient groups, biopharmaceutical companies and scientists from academia, the US FDA, the EMA, the National Institute of Neurological Disorders and Stroke (NINDS), and the National Institute on Aging (NIA). The data available in the CAMD database have been volunteered by CAMD member companies and nonmember organizations. The use of this dataset for our research was approved by the Human Research Ethics Committee at The University of Newcastle, Australia (approval number: H-2014-0335).

The CAMD dataset contains, among others, patients' demographic, cognitive test result and *APOE* genotype data, and information about conducted medication trials. The MMSE test records consist of six entries each: five corresponding to the results in questionnaire categories orientation, registration, attention, recall and language, and the sixth to the total score (where a nonstandard MMSE test was applied, the final score was normalized to a 30-points scale). The data preprocessing step, employed to normalize the different clinical trials in the CAMD database, is explained in more detail in Subsection MMSE test results.

For approximately a third of all patients, the information about their *APOE* genotype status is available. These patients are assigned with two alleles, each of the type e2, e3 or e4.

During the medication trials recorded in the CAMD study, patients were given AD-related medications or placebos; intakes of prescribed drugs uncorrelated to AD were also registered. Due to variations among the trials and a wide range of medications, a preprocessing of medication data was necessary to statistically analyze their effect; this procedure is described in Subsection Medications intake.

Data preprocessing

The original CAMD data were collected by several studies employing varying standards. Therefore, this database required preprocessing, which is described in this section. We would like to emphasize that the procedures outlined in Subsections MMSE test results and Medications intake are specific for this particular database and they were applied to enable a normalization across clinical trials and their merging to a single dataset for our clustering analysis purposes; these steps are not part of the actual data analysis.

MMSE test results

In the CAMD dataset, different versions of the MMSE test were applied, under employment of the

same test categories, but varying weights. For our clustering purposes, a normalized multivariate dataset is required, and thus, we needed to utilize normalized scores achieved in each of the test categories orientation, registration, attention, recall and language. To that end, we needed to identify the maximal possible scores that could be obtained in each single cognitive test part. Since this information was not included in this dataset, we applied an iterative approach exploiting score distributions to determine them; details on this procedure are provided in Supplementary Data A. Accordingly, we were able to reliably identify three MMSE scoring systems for a total of 3717 patients. One of the versions corresponds to the standard interpretation employing the 30-point scoring system with 10, 3, 5, 3 and 9 points assigned to the categories orientation, registration, attention, recall and language, respectively. Some patients were tested using a 56-point system, with the maximal scores of 20, 6, 10, 6 and 14 in the five categories, respectively. These values correspond to the relative contributions of 35.7, 10.7, 17.9, 10.7 and 25%; this is consistent with the standard version, except the language category is attributed less significance. Another 51-point scoring system, also employed in the CAMD study, is based on the maximal achievable points of 20, 6, 5, 6 and 14 in the same categories, corresponding to the relative weights of 39.2, 11.8, 9.8, 11.8 and 27.5%, respectively. In this version, the categories orientation, registration and recall have a greater impact on the total single score than in the standard. The identification of these weighting schemes was employed to enable a normalization of the results across different trials, as further discussed in Subsection MMSE test results normalization.

Medications intake

There are 60,708 records in the medications domain, corresponding to 12,346 different drug IDs, provided for 5996 patients. This is a very sparse data, with a cov-

erage of just $0.082\% \left(\frac{60,708}{5996 \cdot 2,346}\right)$. To be able to draw information from this domain, a preprocessing including data merging was necessary. For these purposes, we needed to disregard the intake dosage, time and duration of each drug treatment, although we understand that this information is not insignificant. We applied a recursive approach aiming at clustering together drugs with same generic but different advertized brand names, based on the category tags they share. This led to a total of 132 medication titles. Within this list, we further identified medications recorded as being related to AD treatment: those targeting to treat dementia of AD type (donepezil, galantamine, rivastigmine and memantine), anti-inflammatory medications (ibuprofen, aspirin, celecoxib and naproxen), schizophrenia-related drugs (risperidone and quetiapine), antidepressants (paroxetine, trazodone, sertraline and citalopram), anti-anxiety medications (lorazepam) and cardiovascular conditions-related treatment (clopidogrel and atorvastatin). The vitamins B6, B9, B12, C, E and multivitamins, fish oil and Ginkgo biloba were used as placebos. More details on the original titles of these medications and placebos are provided in Supplementary Data B.

MMSE test results normalization

Application of the data preprocessing step led to a dataset containing MMSE test results stratified by categories, and the corresponding scoring systems utilized to measure them for each of the 3717 patients. We further used this information to normalize the CAMD dataset across different MMSE test versions, in order to enable the employment of all samples in a single cohort. Thus, we transformed the number of points in each MMSE test category to percentages, with relation to the corresponding maximal possible scores. Accordingly, all normalized values lie between 0 and 100%, where 0% stands for 0 points, and 100% for the maximal score. For example, if each of the patients A and B achieved ten points in the category orientation, where the former patient was examined using the 30-point scoring system (maximal ten points for this test category), and the latter 56-point system (maximal 20 points), then although their absolute scores are the same, the performance of the patient A is 100%, while it is only 50% of the patient B. These are the success rates we considered for further data analysis purposes.

Clustering of AD patients

In this study, we aim at identifying naturally emerging groups of AD patients based on their multivariate cognitive test results and building a classification model for their determination. Therefore, we first randomly subdivided the cohort comprising 3717 patients into a training (1858 samples) and a validation (1859 samples) set. We then applied a clustering approach to the training set; since the number of features is low (five test categories), we employed the ordinary Euclidean metric to calculate the distance between samples, in accordance with Euclidean distance calculation, Equation 1, where d(p,q) stands for distance between multivariate results of samples p and q, *i* indicates the test category and N indicates their total set (five in our case). Each value p and q ranges between 0 and 100%, and thus, the distance values d(p,q) lie between 0% and $\sqrt{(5.100\%)^2}$ ($\approx 223.6\%$).

$$d(\mathbf{p},\mathbf{q}) = \sqrt{\sum_{i \in N} (q_i - p_i)^2}$$

Equation 1

The distance matrix containing Euclidean metric values between all sample pairs in the training set was further employed to compute the Ward hierarchical clustering, where the variance within each group of samples is minimized [31], using R [32]. This calculation led to clusters of patients characterized by mutual cognitive test outcomes; the number of clusters was determined using a trade-off between minimizing this value and maximizing their differentiation from each other.

Since each cognitive MMSE test was conducted and repeated on different dates for same individuals, for the clustering purposes in this study, we utilized records from the first visit date measured before the implementation of medication trials.

Signature & validation of AD patient groups

Once the patient groups had been established, we aimed at identifying test categories contributing to their separation the most. Thus, we applied the multidimensional rank-based Kruskal–Wallis test ('one-way ANOVA on ranks') [33] to each category separately to define its segregation power, using R. The subset of categories with significantly low p-values was further used for centroid calculation for each patient group in the training set; these are the signature vectors containing mean scores of each relevant category for each patient group.

Samples in the validation set, also corresponding to the first cognitive test visit dates, were assigned to one of the patient groups based on their Euclidean distance (Equation 1) to the closest centroid calculated above.

Transitions across patient groups over time

To analyze patient transitions across the groups, we considered the first and last MMSE test results: the former corresponding to any time before or at the beginning of AD medication trials, and the latter at the end or after. Time intervals between the emerging cognitive test results strongly vary across samples with some extending up to 6.5 years. We selected the last visit date to lie within a time period between 6 months and 4 years after the first visit; this choice corresponds to a trade-off between minimizing the time frame between the tests and maximizing the number of samples, while keeping a minimal time lapse between the tests, in accordance with the histogram shown in Supplementary Data C. We found that 1197 samples in the training and 1172 samples in the validation set fulfilled this criterion.

The assignment of patients to a group based on their MMSE test results achieved at the last visit in both training and validation sets was conducted using centroids previously calculated based on the training set (subsection signature & validation of AD patient groups), by means of the Euclidean distance (Equation 1).

Statistical tests

We computed the percentage agreement and Fleiss' Kappa κ [34], a statistic measure for assessing the reliability of this agreement, between the patient group labels in the training set emerging from the previously described clustering procedure and those assigned using centroids based on the categories defined to be relevant for these groups segregation. We further computed the same measures for labels assigned in the clustering process and those based on centroids calculated for all five cognitive test categories. The objective of this analysis is to provide insights about the performance of our mathematical model using centroids to represent the resulting patient groups.

We employed the *APOE* genotype information to analyze the association between this marker and patient groups. To that end, we applied the proportion test examining the null hypothesis that the proportions of certain alleles in several groups of patients are the same [35].

We used the medication domain to correlate group transitions over time with certain treatment types. Since AD is generally considered a one-directional disease and patients' cognitive state generally worsens over time, we compared intake incidence rates of medications in one transition group to those corresponding to a relatively better or worse change. Titles that were ingested by more patients in a transition associated with less severe changes are considered to have a beneficial effect, and vice versa. To calculate significance values of these correlations, we used the binomial test; this is the 2D version of the proportion test. All statistical tests were performed using R [32].

Results & discussion

Hierarchical clustering of cognitive test results determines five groups of patients

The clustering approach applied to the training set led to five groups of AD-diagnosed patients, as shown in Figure 1A. For their computation, we employed patients' success rates ranging between 0 and 100% and representing normalized original absolute scores assigned to each category separately (e.g., a 100% success in the category orientation corresponds to 10 points in the standard MMSE test, while a 50% rate is equal to 5 points in the same test and category). As shown in this figure, the emerging patient groups are associated with divergent incidence rates: cognisant (194 samples), inattentive (261), forgetful (632), distant (589) and absent (182). The relatively small cognisant group is characterized by an overall high performance, particularly in the test categories registration, attention and language; patients in this group only show limited impairment in orientation and recalling abilities. The inattentive group has difficulties with attention and exhibits slight alterations in orientation and recalling skills. The forgetful patients, constituting the largest group, are unable to remember and show slight disorientation. The distant group is characterized by a degradation of patients' attention and recalling abilities, an impairment in orientation and a slight deterioration of linguistic skills. Patients of the last and smallest group, the absent, are unable to succeed in any of the MMSE test categories.

To examine which categories significantly contribute to the definition of the five patient groups, we applied the Kruskal-Wallis test to each category separately. The categories registration, attention and recall showed a very strong association with the groups stratification (p-values of 4.3×10^{-302} , 2×10^{-294} and 4.7×10^{-270} , respectively). Although the remaining two categories were still associated with significant results, their log₁₀normalized p-values were almost threefold smaller (p-values of 9.6×10^{-117} for orientation and 1×10^{-100} for language). Thus, we selected the three categories with the smallest p-values to build a spanning set of features defining the groups, resulting in a reduction of the problem dimensionality from five down to three (categories). It is also visible from Figure 1A that the language and orientation categories do not define the boundaries between the patient groups as accurate as the registration, attention and recall.

Subsequently, we calculated a centroid for each patient group in the training set based on the selected categories registration, attention and recall. The corresponding values are listed in Table 1. These success rates also represent the previous description of each group, where the cognisant patients perform best in all categories, inattentive tend to show little attention, forgetful are not able to recall, distant are only able to register and absent are unable to succeed in any of these categories.

We further compared the assignment of labels using the reduced dimensionality space to those employing all five categories. The labels agreement between the original assignment and those using centroids calculated for the three selected categories (Table 1) is 92.9% (1726 out of 1858 samples) and Fleiss' Kappa κ is equal to 0.904. An analogous comparison between the patient group labels assigned in the hierarchical clustering process and using centroids calculated based on all five categories led to an agreement value of 90% (1673 out of 1858) and κ equal to 0.866. According to



Figure 1. Heat maps of the training and validation sets. These heat maps show patients clustered into the groups cognisant (yellow), inattentive (green), forgetful (turquoise), distant (red) and absent (blue), and their corresponding MMSE test results achieved in each category (orientation, registration, attention, recall and language). All scores (denoted as 'success rate') were normalized across categories, ranging between 0% (blue) and 100% (red) each, where 0% corresponds to 0 points and 100% to the maximal possible score. The categories significantly differentiating in their success rate values between the patient groups, which were also used for centroids calculation, are denoted with a '*'. (A) Hierarchical clustering of samples based on the outcomes in five cognitive test categories registration, attention and recall. MMSE: Mini-mental state examination.

Landis and Koch [36], the greater the κ values the better the agreement, where values greater than 0.81 stand for an 'almost perfect agreement'. Remarkably, the results in this study do not only confirm the utility of the categories registration, attention and recall for describing the 5D test category space but also demonstrate that they are better representatives. Thus, the five patient groups can be determined in absence of the categories orientation and language, indicating that their supplementary employment along with the three selected test categories does not significantly contribute to the AD group discrimination. Samples in the validation set were assigned to a cognitive patient group in accordance with the closest centroid from Table 1. The corresponding heat map including values of the two MMSE test categories orientation and language that were not utilized in the appointment process, is shown in Figure 1B. There are 246 cognisant, 245 inattentive, 677 forgetful, 511 distant and 180 absent patients in the validation set. The group sizes and heat maps are consistent across both datasets – an observation that supports the dimensionality reduction down to three categories, and our centroids and groups definition.

Table 1. Centroids calculated in the training set.						
Category/patient group	Cognisant (%)	Inattentive (%)	Forgetful (%)	Distant (%)	Absent (%)	
Registration	100	93.3	93.3	100	43.3	
Attention	98	44	94	21	18	
Recall	83.3	83.3	13.3	6.7	0	

Mean success rate values ranging between 0 and 100% in the mini-mental state examination test categories, registration, attention and recall, calculated for the patient groups, cognisant, inattentive, forgetful, distant and absent in the training set.

Patient groups are associated with MMSE total score & APOE genotype

To visualize the relationship between patients' original total MMSE scores and their stratification into the five groups defined above, we generated box plots, as shown in Figure 2A & B for the training and validation sets, respectively. Both datasets show coherent results, where the cognisant patients achieve the most (mean: 26.9 and 26.7 points, respectively) and absent the least scores (mean: 10 and 9.3 points, respectively). The inattentive and forgetful patients share the same mean score of 22.4 points in the training set, and 22 and 21.9 in the validation set. Patients from the distant group achieve on average 16.4 points in both datasets, and thus on the MMSE scores scale, they are located between the absent and all other

groups. Very low p-values (9.7×10^{-255}) in the training and 4.2×10^{-242} in the validation set) reflect the high significance of the separation between the MMSE score distributions across the patient groups defined above; this is also visible from the first and third quartiles of each box plot with no overlaps between them (except the inattentive and forgetful groups with virtually the same scores). Score ranges between the whiskers, however, are large and substantially intersect each other. This means that although there is a clear association between the groups identified in this study and their total MMSE scores originally calculated as the sum of points achieved in all five test categories, these definitions are not the same and there is a substantial part of patients for which this association is void.



Figure 2. Association between the patient groups and mini-mental state examination total scores and APOE

genotypes. (A & B) These box plots show the MMSE total score distributions in the training and validation sets for the five patient groups defined in this study: cognisant (yellow), inattentive (green), forgetful (turquoise), distant (red) and absent (blue). (C & D) These two bar charts display the distributions of patient groups stratified by the *APOE* genotype, in the training and validation sets. The last bar corresponds to the global distribution regardless of the *APOE* genotype. Numbers in parenthesis represent the population size of each stratified group. MMSE: Mini-mental state examination.

We further analyzed the APOE genotypes, a marker widely associated with AD, in relation to the patient groups defined in this study. The resulting bar charts are shown in Figure 2C & D; each bar represents a genotype (APOE e2/e3, e2/e4, e3/e3, e3/e4 or e4/e4) with the corresponding incidence rates. Interestingly, among the AD-diagnosed patients analyzed in this study, the genotype APOE e2/e2 is not represented at all, and e3/e4 is the most common form of APOE; this prevalence rates are divergent from those of a general population [10,37]. As shown in the figure, the genotype APOE e2/e3 contains the largest proportion of cognisant patients, while e4/e4 is the least, in both datasets. Generally, the proportions of cognisant, inattentive and forgetful patients decrease, and the ratios of distant samples increment, in the following order: e2/e3, e3/e3, e3/e4, e4/e4. Group distributions corresponding to the genotype e2/e4 are based on a small number of observations, and thus they are not very reliable; the same holds for the distant group. Since APOE e3/e4 and e4/e4 genotypes can be associated with patients showing poor performance in the MMSE tests, we compared the incidence rates of these two genotypes combined together across our patient groups. The proportion test p-values were found to be small (0.06 in the training and 9.9×10^{-6} in the validation set, computed based on 355 and 338 samples, respectively), suggesting that the difference in genotype distributions is significant. The cognisant and inattentive patients were found to contain the smallest rates of APOE e3/e4 or e4/e4, and the distant the highest. The absent group also holds a great proportion of these two genotypes, however, the population sizes are too small (eight and nine samples in the training and validation sets, respectively) to draw a conclusion. Summarizing, the APOE genotypes show a certain association with the five patient groups defined in this study, however, this relation is rather weak and alone is not enough to predict or determine the patient groups. Instead, the

APOE could be interpreted as a factor supporting or opposing patients' cognitive impairment.

Consistent transitions across patient groups

Transitions across patient groups over a time frame between 6 months and 4 years are captured in Tables 2 & 3 for the training and validation sets, respectively. Both sets show a remarkable similarity with regards to the relative changes between the groups (denoted in parenthesis). A visualization of these changes over time in the training set is additionally shown in Figure 3. This figure can be interpreted as the evolution of patients diagnosed with AD, generally undergoing a medication or placebo trial, during the time period of up to 4 years.

Most patients tend to retain their group association labels within 4 years. Nevertheless, there are differences in transition behavior between the five patient groups. For instance, the forgetful, distant and absent patients mostly keep their status (patient proportions of 52, 57 and 58% in the training set, respectively), while the majority of the inattentive transit to other groups (only 38% of these patients remain in the same group in the training set). The absent group corresponding to the worst performance in all MMSE test categories only shows substantial changes toward the distant (28%), indicating a slight improvement of the cognitive state in terms of the registration abilities. The distant patients, on the other hand, show some transitions into the forgetful (15% of these patients improve with regards to attention) or absent (18% of these patients deteriorate with respect to registration) groups. Some of the forgetful patients become cognisant again (15%) or decline to distant (20%), which is associated with a loss of the capacity for attention. A substantial part of the cognisant patients changes their label to either forgetful (24%) or inattentive (25%), both representing a slight decline: one with respect to the recall and the other one to the attention abilities. Approximately a

Table 2. Transitions across patient groups in the training set.							
First/last	Cognisant	Inattentive	Forgetful	Distant	Absent	Overall	
Cognisant	60 (43)	35 (25)	33 (24)	9 (7)	1 (1)	138	
Inattentive	33 (19)	68 (38)	22 (12)	44 (25)	10 (6)	177	
Forgetful	65 (15)	28 (7)	218 (52)	86 (20)	24 (6)	421	
Distant	13 (3)	25 (7)	57 (15)	218 (57)	67 (18)	380	
Absent	1 (1)	4 (5)	6 (7)	23 (28)	47 (58)	81	
Overall	172	160	336	380	149	1197	

The absolute and relative numbers of patients changing their group label within a time frame between 6 months and 4 years in the training set are listed in this table. Labels assigned to samples based on the MMSE test results achieved at the first visit are represented by rows, and those at the last by columns. Numbers in parenthesis represent the relative number of patients (proportion) involved in each transition, calculated with respect to the whole population corresponding to the label determined at the first visit (last column). The population size of each transition group used for this analysis is listed in the last column for the first visit and in the last row for the last visit.

Table 3. Transitions across patient groups in the validation set.						
First/last	Cognisant	Inattentive	Forgetful	Distant	Absent	Overall
Cognisant	76 (44)	53 (31)	30 (17)	9 (5)	6 (3)	174
Inattentive	35 (20)	56 (32)	19 (11)	48 (27)	18 (10)	176
Forgetful	63 (15)	28 (7)	225 (52)	89 (21)	25 (6)	430
Distant	18 (3)	23 (7)	35 (11)	191 (60)	59 (19)	316
Absent	1 (1)	5 (7)	5 (7)	25 (33)	40 (53)	76
Overall	183	165	314	362	148	1172

The absolute and relative numbers of patients changing their group label within a time frame between 6 months and 4 years in the validation set are listed in this table. Labels assigned to samples based on the MMSE test results achieved at the first visit are represented by rows, and those at the last by columns. Numbers in parenthesis represent the relative number of patients (proportion) involved in each transition, calculated with respect to the whole population corresponding to the label determined at the first visit (last column). The population size of each transition group used for this analysis is listed in the last column for the first visit and in the last row for the last visit.

fifth of the inattentive patients become cognisant again (19%), and a quarter deteriorate to distant by additionally losing the capability for recalling (25%). Although the forgetful and inattentive groups share virtually the same MMSE total scores, they are rather related to each other through their origin from the cognisant group and a common deterioration to the distant; they show only little direct connection.

We also examined whether these transitions can be linked to the *APOE* genotypes. Thus, we applied the proportion test to genotype distributions with regards to the transition groups discussed above. We found no significance in differentiation between these markers; however, these results may be inconclusive due to small population sizes.

Medications associated with transitions across groups

We further analyzed transitions between patient groups within the time frame between 6 months and 4 years on their possible associations with the treatment recorded in the CAMD dataset. From our processed medication list, only three titles were found to be significantly (p-values below 0.05) associated with changes in both the training and validation sets: multivitamins (originally classified as placebo), citalopram (an antidepressant) and ginkgo biloba (placebo). An overview of these correlations is shown in Table 4 and more details are provided in Supplementary Data D.

According to these results, the cognisant patients remaining in the same group for up to 4 years are significantly associated with increased intake incidence rates of multivitamins, when compared with patients transitioning from this group to all others combined together. This indicates that these nutrients may have a positive effect and potentially support the ADdiagnosed patients in keeping their cognisant status, since a transition from this group to any other group can be considered as a cognitive deterioration. Recent studies on the vitamins B family, C, D and E also support this finding by emphasizing that they help maintaining a healthy neuronal population and delay brain aging associated with the onset and mild AD [38-42].

A deterioration from the inattentive group toward absent is significantly correlated with increased intake incidence rates of citalopram. Thus, this medication is considered to have an adverse effect, since the absent group is considered to represent a worse cognitive impairment state than inattentive. Previous studies have also shown that although citalopram helps reducing agitation in AD, its practical application is limited by an induced patients' cognitive decline [43].



Figure 3. Transitions across patient groups in the training set. Patient transitions between the five cognitive groups in the training set are visualized in this graph. The changes were captured using the time frame between six months and four years. The arrow line thickness is proportional to the fraction of people in each transition (the wider the greater is the proportion); only rates greater than 10% are captured in this graph. Arrows leading to the same group stand for patients retaining their status.

Table 4. Details on medications associated with transitions across patient groups.						
From	То	Treatment	Effect	p-value (T)	p-value (V)	
Cognisant	Cognisant	Multivitamins	Beneficial	0.031	0.025	
Inattentive	Absent	Citalopram	Adverse	0.0056	0.031	
Distant	Absent	Ginkgo biloba	Adverse	0.019	0.046	

Significant associations between increased medication intake incidence rates and transitions between patient groups ('From' and 'To') over the time frame of up to 4 years are listed in this table. Medication titles are listed in the column 'Treatment'. 'Effect' indicates the relative impact of the corresponding drug on the original patient group depicted in the column 'From'; 'beneficial' means that the intake of a particular drug listed in 'Treatment' is associated with relatively positive changes for the patient group listed in the column 'From'; 'beneficial' means that the intake of a particular drug listed in 'Treatment' is associated with relatively positive changes for the patient group listed in the column 'From', and 'adverse' stands for a negative effect and indicates that the corresponding treatment should be prescribed with caution for patients in the group listed in the column 'From'. The Binomial test p-values are recorded in the last two columns for the training ('T') and validation ('V') (atasets, respectively; more details on the calculation of these values are provided in **Supplementary Data D**.

Our results further demonstrate that Ginkgo biloba may have a negative effect on distant patients, due to its increased intake incidence rates being associated with transition to the absent group. Its efficacy has been discussed in the literature before, but remains elusive [44-46]. It is also possible that divergent outcomes could be linked to varying mental states of AD patients ingesting this medicinal herb.

Although we were able to depict several associations between the group transitions and medication intakes, many drug titles could not be significantly correlated to a cognitive state improvement or deterioration due to data sparsity. Furthermore, we would like to emphasize that the results of this analysis are only indicative, due to the lack of consistency in the medication trials comprising treatments of varying types and lengths.

Conclusion & implications

Patient groups: cognitive subtypes or disease progression?

Clustering of patients based on their cognitive test results in the categories orientation, registration, attention, recall and language, led to an identification of five naturally emerging groups: cognisant, inattentive, forgetful, distant and absent. These patient clusters showed distinct and consistent characteristics in terms of the population sizes and total MMSE scores, and transitions across them indicated that these entities can be associated with varying prognosis. It is inconclusive, however, whether these patient groups represent intrinsic cognitive subtypes of AD or disease progression.

Patient groups defined through the naturally emerging clusters may be interpreted as the representatives for most common and varying multivariate cognitive MMSE test outcomes. This constellation alone does not imply that there are no continuous changes between these groups; it is possible that the clusters themselves could express a progression from one to another. Nevertheless, the varying cluster sizes indicate that there are underlying differences in the occurrence rates within the AD-diagnosed patients: the cognisant and absent patients are substantially less common than inattentive and distant.

In terms of progression paths, there are at least two possible alternatives, also evident in Figure 3: first, from cognisant via forgetful and subsequently distant to absent and second, from cognisant via inattentive and subsequently distant to absent. In the first path, patients lose the ability to remember (forgetful), followed by an additional incapacity for attention (distant) and a decline to the final stage where they are unable to register events (absent). The second path is delineated by loss of patients' attention (inattentive), followed by an additional incapability for recall (distant) and a further deterioration denoted by the loss of registration skills (absent). Remarkably, following this hypothesis, the forgetful and inattentive patient groups represent different paths with no direct connections between each other (they can be linked to each other using the passage through the cognisant or distant states).

Under consideration of AD being a progressive disease delineated by continuously declining patients' cognitive states, we suggest the perception of each alternative path described above as a disease progression course from the best to the worst phase, where the most common phases are depicted by our patient clusters. The groups forgetful and inattentive, however, could be considered as cognitive subtypes of this disease. Remarkably, both of these groups show virtually the same MMSE total scores and thus are undetectable using the commonly employed total score as the sum of all points achieved in the five test categories.

MMSE test categories space reduction & patients' educational background

Along with the five patient groups, we also identified three MMSE test categories, out of five, mostly contributing to their separation: registration, attention and recall. These are the features that are also sufficient to represent the patient groups. Interestingly, the three selected cognitive test categories can be regarded as examination of basic skills unrelated to further qualifications including the language, which is affected by the education level variations across patients the most. These insights may lead to an improvement of the cognitive test design and also resolve the issue of the educational background, which is currently leading to varying test standards in attempt to adjust the resulting total MMSE scores.

Translation into practice

The results of our study suggest that the currently employed cognitive test categories registration, attention and recall can be used to stratify AD-diagnosed patients into five groups: cognisant, inattentive, forgetful, distant and absent. This is with the purpose of defining the disease progression state and providing prognosis for patients based on data listed in Tables 2 & 3 and visualized in Figure 3. Following the requisite for translation of our methodology into practice, we describe how the centroids from Table 1 can be applied in clinics in more details in Supplementary Data E, and we provide an Excel file with an implemented calculator in Supplementary Data F & Supplementary Table 1.

Implications & future outlook

By conducting a multivariate analysis of AD-diagnosed patients, we revealed new insights about the most common AD patient groups and associated relevant cognitive test categories (section 'Results & discussion'); we also built a model to assign each AD patient to one of these groups in clinics (Table 1 & Supplementary Data F). While the naturally emerged advantage of our centroids is the absence of the categories orientation and language, which are affected by patients' varying educational backgrounds the most, the main benefit of this application is its independence of the single weights assigned to each test category. This means that any nonstandard cognitive test containing the categories registration, attention and recall can be employed to examine patients' state, while the interpretation of the results will remain the same. This allows an immediate prognosis in accordance with the results obtained in this study, and a direct comparison between patients across different studies, unbiased by the cognitive space reduction from five down to one single total MMSE score.

We discussed in subsection patient groups: cognitive subtypes or disease progression? that the forgetful and inattentive patients may represent divergent progression paths. A further investigation is required to exclude a bias arising from a possible misdiagnosis and provide more understanding about their underlying nature.

The analysis of the scarce treatment data (subsection medications associated with transitions across groups) pointed to the beneficial effects of multivitamins for the cognisant group. Citalopram showed an adverse effect on the inattentive and Ginkgo biloba on the distant patients, where increased intake incidence rates are associated with a deterioration to absent. We found no significant improvement, however, in individuals administered with the remaining AD-specific drugs or placebos assayed in the CAMD trials. While this result may be biased by the sparsity of the numbers of observations and variations in prescription details, it still shows that AD remains an incurable one-directional disease, where the most positive transitions associated with a treatment are represented by patients retaining their status for up to 4 years without declining.

Future perspective

We believe that in 5-10 years' time, the multivariate analysis approaches will take overhand in clinical applications, including the AD impairment state assessment procedure. Thus, we believe that our results may represent the first steps toward this goal, where we demonstrate that AD patients may be stratified into five distinct groups based on their multivariate cognitive test outcomes. More importantly, our analysis led to the conclusion that the category language is not required for these patient groups determination, given that the categories registration, attention and recall are measured. Thus, an exclusion of this part of the test may lead to an improvement of the MMSE test interpretation in the future, since this category is potentially affected by patients' educational background the most.

Supplementary

To view the supplementary data that accompany this paper please visit the journal website at: www.future-science.com/ doi/full/10.4155/fsoa-2016-0041

Acknowledgements

P Moscato expresses his gratitude to the anonymous donors for their support of his Alzheimer's Disease research program.

Disclaimer

This study makes use of the Coalition Against Major Diseases (CAMD) database. Data used in the preparation of this article were obtained from the CAMD database. As such, the investigators within CAMD contributed to the design and implementation of the CAMD database and/or provided data, but did not participate in the analysis of the data or the writing of this report.

Financial & competing interests disclosure

P Moscato also acknowledges funding of his bioinformatics research from the Australian Research Council (ARC, www. arc.gov.au/) Future Fellowship FT120100060. The authors have no other relevant affiliations or financial involvement with any organization or entity with a financial interest in or financial conflict with the subject matter or materials discussed in the manuscript apart from those disclosed.

No writing assistance was utilized in the production of this manuscript.

Ethical conduct of research

The authors state that they have obtained appropriate institutional review board approval or have followed the principles

outlined in the Declaration of Helsinki for all human or animal experimental investigations. In addition, for investigations involving human subjects, informed consent has been obtained from the participants involved.

Open access

This work is licensed under the Creative Commons Attribution 4.0 License. To view a copy of this license, visit http://creativecommons.org/licenses/by/4.0/

Executive summary

- The mini-mental state examination (MMSE) is a questionnaire employed to assess the Alzheimer's disease (AD)-diagnosed patients' mental state; it consists of five test categories: orientation, registration, attention, recall and language.
- Evaluation of the MMSE outcomes by means of a single total score is biased by the patients' educational background and varying test versions employing the same categories, but different relative weight attributions.
- The multivariate analysis on cognitive test outcomes conducted in this study led to an identification of five groups of AD patients: cognisant, inattentive, forgetful, distant and absent.
- These patient groups are associated with distinct characteristics and prognostics.
- Out of the five MMSE questionnaire categories, only three were found to be critical for these patient groups determination: registration, attention and recall.
- The mathematical model (centroids) proposed in this study for identification and characterization of the AD patient groups is robust, it may help resolving the patients' educational background bias and it allows a direct comparison across patients independent of the cognitive test version (provided it contains the categories registration, attention and recall).
- Multivitamins may support cognisant AD-diagnosed patients with a mild recalling impairment.

References

Papers of special note have been highlighted as: • of interest; •• of considerable interest

- 1 Alzheimer's Association. 2015 Alzheimer's disease facts and figures. *Alzheimers Dement.* 11(3), 332–384 (2015).
- Provides a comprehensive overview about Alzheimer's disease.
- 2 Molsa PK, Marttila RJ, Rinne UK. Survival and cause of death in Alzheimer's disease and multi-infarct dementia. *Acta Neurol. Scand.* 74(2), 103–107 (1986).
- 3 Molsa PK, Marttila RJ, Rinne UK. Long-term survival and predictors of mortality in Alzheimer's disease and multi-infarct dementia. *Acta Neurol. Scand.* 91(3), 159–164 (1995).
- 4 Jack CR Jr, Albert MS, Knopman DS et al. Introduction to the recommendations from the National Institute on Aging-Alzheimer's Association workgroups on diagnostic guidelines for Alzheimer's disease. *Alzheimers Dement.* 7(3), 257–262 (2011).
- 5 Albert MS, Dekosky ST, Dickson D *et al.* The diagnosis of mild cognitive impairment due to Alzheimer's disease: recommendations from the National Institute on Aging-Alzheimer's Association workgroups on diagnostic guidelines for Alzheimer's disease. *Alzheimers Dement.* 7(3), 270–279 (2011).
- 6 Sperling RA, Aisen PS, Beckett LA *et al.* Toward defining the preclinical stages of Alzheimer's disease: recommendations from the National Institute on Aging-Alzheimer's Association workgroups on diagnostic guidelines for Alzheimer's disease. *Alzheimers Dement.* 7(3), 280–292 (2011).

- 7 Krueger KR, Wilson RS, Kamenetsky JM, Barnes LL, Bienias JL, Bennett DA. Social engagement and cognitive function in old age. *Exp. Aging Res.* 35(1), 45–60 (2009).
- 8 Wang HX, Xu W, Pei JJ. Leisure activities, cognition and dementia. *Biochim. Biophys. Acta* 1822(3), 482–491 (2012).
- 9 Stern Y. Cognitive reserve in ageing and Alzheimer's disease. *Lancet Neurol.* 11(11), 1006–1012 (2012).
- 10 Farrer LA, Cupples LA, Haines JL *et al.* Effects of age, sex, and ethnicity on the association between apolipoprotein E genotype and Alzheimer disease. A meta-analysis. APOE and Alzheimer Disease Meta Analysis Consortium. *JAMA* 278(16), 1349–1356 (1997).
- 11 Reitz C, Jun G, Naj A *et al.* Variants in the ATP-binding cassette transporter (ABCA7), apolipoprotein E 4, and the risk of late-onset Alzheimer disease in African Americans. *JAMA* 309(14), 1483–1492 (2013).
- 12 Liu CC, Kanekiyo T, Xu H, Bu G. Apolipoprotein E and Alzheimer disease: risk, mechanisms and therapy. *Nat. Rev. Neurol.* 9(2), 106–118 (2013).
- 13 Waldemar G, Dubois B, Emre M *et al.* Recommendations for the diagnosis and management of Alzheimer's disease and other disorders associated with dementia: EFNS guideline. *Eur. J. Neurol.* 14(1), e1–e26 (2007).
- 14 Tombaugh TN, McIntyre NJ. The mini-mental state examination: a comprehensive review. *J. Am. Geriatr. Soc.* 40(9), 922–935 (1992).
- 15 Pangman VC, Sloan J, Guse L. An examination of psychometric properties of the mini-mental state examination

and the standardized mini-mental state examination: implications for clinical practice. *Appl. Nurs. Res.* 13(4), 209–213 (2000).

- 16 Folstein MF, Folstein SE, McHugh PR. "Mini-mental state". A practical method for grading the cognitive state of patients for the clinician. *J. Psychiatr. Res.* 12(3), 189–198 (1975).
- •• Contains the original version of the mini-mental state examination (MMSE) test.
- 17 Folstein MF, Folstein SE. *Mini-Mental State Examination (2nd Edition), (MMSE-2).* PAR Inc, FL, USA (2010).
- The current version of the MMSE test translated in different languages is provided in this documentation.
- 18 Rovner BW, Folstein MF. Mini-mental state exam in clinical practice. *Hosp. Pract.* 22(1A), 99, 103, 106, 110 (1987).
- Mungas D. In-office mental status testing: a practical guide. *Geriatrics* 46(7), 54–58, 63, 66 (1991).
- 20 Strauss E, Sherman EMS, Spreen O, Spreen O. A Compendium of Neuropsychological Tests: Administration, Norms, and Commentary (3rd Edition). Oxford University Press Inc., New York, USA (2006).
- 21 Loewenstein DA, Barker WW, Harwood DG *et al.* Utility of a modified mini-mental state examination with extended delayed recall in screening for mild cognitive impairment and dementia among community dwelling elders. *Int. J. Geriatr. Psychiatry* 15(5), 434–440 (2000).
- 22 Newkirk LA, Kim JM, Thompson JM, Tinklenberg JR, Yesavage JA, Taylor JL. Validation of a 26-point telephone version of the mini-mental state examination. *J. Geriatr. Psychiatry Neurol.* 17(2), 81–87 (2004).
- 23 Teng EL, Chui HC. The modified mini-mental state (3MS) examination. J. Clin. Psychiatry 48(8), 314–318 (1987).
- 24 Crum RM, Anthony JC, Bassett SS, Folstein MF. Populationbased norms for the mini-mental state examination by age and educational level. *JAMA* 269(18), 2386–2391 (1993).
- The authors of this article discuss how the educational background can affect the MMSE in total scores.
- 25 Gugssa SA, Davey G, Ejigu AA, Metaferia GZ, Medhin G, Kelkile TS. Population norms for the mini-mental state examination in Ethiopia. *Ethiop. Med. J.* 49(3), 239–247 (2011).
- 26 Bravo G, Hebert R. Age- and education-specific reference values for the mini-mental and modified mini-mental state examinations derived from a non-demented elderly population. *Int. J. Geriatr. Psychiatry* 12(10), 1008–1018 (1997).
- 27 Brito-Marques PR, Cabral-Filho JE. Influence of age and schooling on the performance in a modified mini-mental state examination version: a study in Brazil northeast. *Arg. Neuropsiquiatr.* 63(3A), 583–587 (2005).
- 28 Moraes C, Pinto JA Jr, Lopes MA, Litvoc J, Bottino CM. Impact of sociodemographic and health variables on minimental state examination in a community-based sample of older people. *Eur. Arch. Psychiatry Clin. Neurosci.* 260(7), 535–542 (2010).
- 29 Wong H, Larre P, Ghawche F. Polynesian adaptation of the mini-mental state examination. *Rev. Neurol. (Paris)* 171(4), 359–366 (2015).

- 30 Romero K, De Mars M, Frank D *et al.* The Coalition Against Major Diseases: developing tools for an integrated drug development process for Alzheimer's and Parkinson's diseases. *Clin. Pharmacol. Ther.* 86(4), 365–367 (2009).
- •• Contains the dataset analyzed in this study.
- 31 Murtagh F, Legendre P. Ward's Hierarchical Agglomerative Clustering Method: which algorithms implement ward's criterion? J. Classif: 31(3), 274–295 (2014).
- 32 R Core Team. R: a language and environment for statistical computing (2015).
 - https://cran.r-project.org/doc/manuals
- 33 Hollander M, Wolfe DA. Nonparametric Statistical Methods. Wiley, New York, USA (1973).
- 34 Gamer M, Lemon J, Singh IFP. IRR: various coefficients of interrater reliability and agreement (2012). https://cran.r-project.org/web/packages
- 35 Newcombe RG. Interval estimation for the difference between independent proportions: comparison of eleven methods. *Stat. Med.* 17(8), 873–890 (1998).
- 36 Landis JR, Koch GG. Measurement of observer agreement for categorical data. *Biometrics* 33(1), 159–174 (1977).
- 37 Singh PP, Singh M, Mastana SS. APOE distribution in world populations with new data from India and the UK. *Ann. Hum. Biol.* 33(3), 279–308 (2006).
- 38 Mohajeri MH, Troesch B, Weber P. Inadequate supply of vitamins and DHA in the elderly: implications for brain aging and Alzheimer-type dementia. *Nutrition* 31(2), 261–275 (2015).
- 39 Cacciapuoti F. Lowering homocysteine levels with folic acid and B-vitamins do not reduce early atherosclerosis, but could interfere with cognitive decline and Alzheimer's disease. J. Thromb. Thrombolysis 36(3), 258–262 (2013).
- 40 Carrasco-Gallardo C, Farias GA, Fuentes P, Crespo F, Maccioni RB. Can nutraceuticals prevent Alzheimer's disease? Potential therapeutic role of a formulation containing shilajit and complex B vitamins. *Arch. Med. Res.* 43(8), 699–704 (2012).
- 41 Douaud G, Refsum H, De Jager CA *et al.* Preventing Alzheimer's disease-related gray matter atrophy by B-vitamin treatment. *Proc. Natl Acad. Sci. USA* 110(23), 9523–9528 (2013).
- Smith AD, Refsum H. Homocysteine, B vitamins, and cognitive impairment. Annu. Rev. Nutr. 36, 211–239 (2016).
- 43 Porsteinsson AP, Drye LT, Pollock BG *et al.* Effect of citalopram on agitation in Alzheimer disease: the CitAD randomized clinical trial. *JAMA* 311(7), 682–691 (2014).
- 44 Shi C, Liu J, Wu F, Yew DT. Ginkgo biloba extract in Alzheimer's disease: from action mechanisms to medical practice. *Int. J. Mol. Sci.* 11(1), 107–123 (2010).
- 45 Vellas B, Coley N, Ousset PJ *et al.* Long-term use of standardised Ginkgo biloba extract for the prevention of Alzheimer's disease (GuidAge): a randomised placebocontrolled trial. *Lancet Neurol.* 11(10), 851–859 (2012).
- 46 Weinmann S, Roll S, Schwarzbach C, Vauth C, Willich SN. Effects of Ginkgo biloba in dementia: systematic review and meta-analysis. *BMC Geriatr.* 10, 14 (2010).